Author’s response to reviews

Title: Diagnostic value of the lumbar extension-loading test in patients with lumbar spinal stenosis: a cross-sectional study

Authors:

Naoto Takahashi (naotot211@aol.com)
Shin-ichi Kikuchi (sinichik@fmu.ac.jp)
Shoji Yabuki (yabuki@fmu.ac.jp)
Koji Otani (kotani@fmu.ac.jp)
Shin-ichi Konno (skonno@fmu.ac.jp)

Version: 4 Date: 28 May 2014

Author’s response to reviews: see over
Reynaldo Aldea, Jr.
Journal Editorial Office
BioMed Central

May 28, 2014

Dear Mr. Aldea,

Thank you for your e-mail. We agree with your suggestion to revise certain parts of our manuscript. We moved the Ethics and consent statement to the Methods section of your manuscript, and also clarify consent statement. We have highlighted all changes in the further revised version with underlined text.

We hope that our manuscript is now acceptable for publication in *BMC Musculoskeletal Disorders*. If you have any questions or require further information, please do not hesitate to contact me. Thank you in advance for your consideration.

Sincerely,

Naoto Takahashi, MD, PhD
Department of Orthopaedic Surgery
Fukushima Medical University School of Medicine
1 Hikarigaoka, Fukushima 960-1295, Japan
Tel: +81-24-547-1276, Fax: +81-24-548-5505, E-mail: naotot211@aol.com